Author's response to reviews

Title: Long term hemodialysis aggravates lipolytic activity reduction and very low density, low density lipoproteins composition in chronic renal failure patients.

Authors:

Khedidja Mekki (khmekki@hotmail.com)
Josiane Prost (jprost@u-bourgogne.fr)
Mustapha Remaoun (mremaoun@hotmail.com)
Jacques Belleville (jbellev@u-bourgogne.fr)
Malika Bouchenak (contact@lncm-univ-dz.com)

Version: 3 Date: 13 August 2009

Author's response to reviews: see over
Dear Professor,

Thank you for considering the publication of our manuscript «Long term hemodialysis aggravates lipolytic activity reduction and very low density, low density lipoproteins composition in chronic renal failure patients» in “BMC Cardiovascular Disorders”

We made changes to the manuscript as it was indicated and in accordance with the journal style.

Yours sincerely
Title- Please spell out VLDL and LDL in full.

VLDL and LDL are shown in the title

Section headings ? Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.

we corrected all headings and sub-headings in the manuscript

Page 2- Please delete the keywords.

key words were deleted from abstract

Introduction - Please rename this section ?Background?.

Introduction is renamed Background

Materials & Methods - Please rename this section 'Methods'.

Materials & Methods section is renamed 'Methods'

Authors' contributions - Please use initials to refer to each author's contribution.

Initials are used to refer to each author's contribution.

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Margins of figures are reduced to a minimum